Author’s response to reviews

Title: Improving Nutrition and Physical Activity Environments of Family Child Care Homes: The rationale, design and study protocol of the ‘Healthy Start/Comienzos Sanos’ cluster randomized trial

Authors:
Patricia M. Risica (Patricia_Risica@brown.edu)
Alison Tovar (Alison_Tovar@uri.edu)
Vanessa Palomo (Vanessa_Palomo@brown.edu)
Laura Dionne (laura_dionne@brown.edu)
Noereem Mena (mena23@my.uri.edu)
Kate Magid (kate_magid@alumni.brown.edu)
Diane Stanton Ward (dsward@email.unc.edu)
Kim Gans (Kim.Gans@uconn.edu)

Version: 1 Date: 14 Dec 2018

Author’s response to reviews:

Technical Comments:
- Please ensure that ‘study protocol’ appears in your manuscript title.

** This is completed
- Please ensure that the trial registration date, registry name and the trial registration number are appended to the Trial Registration section at the end of the Abstract in the online submission field, and in the manuscript.

** This is completed
- Please place the Declarations and Abbreviations sections in this order following the Discussion and prior to the References.

** This is completed
- Please remove the Clinical Trial registration information from the Declarations.

** This is completed
- Please state the role of the funding body in the design of the study and collection, analysis, and
interpretation of data and in writing the manuscript. This should be declared in the funding section of the Declarations.

** This is completed
- According to the ICMJE guidelines, to qualify as an author one should have:
  - made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; AND
  - been involved in drafting the manuscript or revising it critically for important intellectual content; AND
  - given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; AND
  - agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Please kindly clarify in the Authors’ contribution section of the Declaration whether each author contributed to other relevant aspects of the study and manuscript preparation. This section should be formatted according to BMC Public Health Guidelines (https://bmcpublichealth.biomedcentral.com/submission-guidelines/preparing-your-manuscript/study-protocol).

If any individual does not qualify for authorship we would ask that they are included in the Acknowledgements only. To request a change in authorship, please download the 'Request for change in authorship form' which can be found here - http://www.biomedcentral.com/about/editorialpolicies#authorship. If relevant, please send the completed form to our editorial office BMCSeriesEditorial@biomedcentral.com.

** This is completed
- If your manuscript does not contain data from any individual person, please state “Not applicable” in the Consent for publication section of the Declarations.

- The main manuscript should be subdivided into the following headings:
  - Background
  - Methods
  - Results
  - Discussion
  - Conclusions

** This is completed - Results section was not included as data are not included in this protocol paper.

- Please remove Figure titles and legends from the Figures, and collate and place Figure Numbers/Titles/Legends in a dedicated section following the references.

** This is completed
- Please remove Figures that are embedded in the manuscript and attach these as separate files.
- Please remove the attached SPIRIT Checklist as this was for our editorial records.

Editor Comments:

- We note that the ethics proof you have provided from The Institutional Review Board of Brown University expired in August 2016. Additionally, you state in the manuscript that Ethics approval was also obtained from The Institutional Review Board of Connecticut University, but this was not included in your submission. BMC Public Health requires ethics approval to have been obtained from all sites where the research is to take place before submission of a study protocol. Before we can proceed with your submission, can you please forward copies of all original and signed documentation that details that ethical approval is currently in place and ongoing at all sites where the study will be conducted. These documents should be sent as email attachments to the following email address, BMCSeriesEditorial@biomedcentral.com. Please DO NOT upload these documents as additional files in the submission system. If your documents are not in English, please could you provide translated versions of the relevant parts. These should be endorsed and signed by a contactable person at the institution. Please also include the original documents. Can you please confirm in your cover letter that you have forwarded the requested documentation to BMCSeriesEditorial@biomedcentral.com.

- The protocol must be for a study that is ongoing. An ‘ongoing’ study is defined as one where the investigators are still collecting, or analyzing data. Can you please confirm and detail the stage at which your study is currently at. Please also confirm that no results from this study have been published, or have been submitted for publication.

- If any of the images in the manuscript are not your own please provide acknowledgement of the source in the figure legend. If any image is copyrighted (for example, a branded product such as the cereal labels in Figure 2), please provide proof of permission from the copyright holder. If the above conditions are not met, the image needs to be removed.

Concerning figures have been changed to provide no branded products.